Author's response to reviews

Title: Tobramycin exposure from active calcium sulfate bone graft substitute

Authors:

Francoise Livio (francoise.livio@chuv.ch)
Peter Wahl (Wahlp@h-fr.ch)
Emmanuel Gautier (GautierE@h-fr.ch)
Chantal Csajka (chantal.csajka@chuv.ch)
Thierry Buclin (thierry.buclin@chuv.ch)

Version: 2 Date: 10 September 2013

Author's response to reviews:

1. Abstract has been revised according to BMC guidelines.
2. Authors' contribution has been included in the manuscript in a section before the Acknowledgements and Reference list.
3. A competing interests section has been included, between the Conclusions and the Authors' contribution ("The authors declare that they have no competing interests").
4. We are waiting for our trial registration number.